

G OPEN ACCESS

Citation: Timmins IR, The PRACTICAL Consortium, Dudbridge F (2024) Bayesian approach to assessing population differences in genetic risk of disease with application to prostate cancer. PLoS Genet 20(4): e1011212. https://doi. org/10.1371/journal.pgen.1011212

Editor: Michael P. Epstein, Emory University, UNITED STATES

Received: December 12, 2022

Accepted: March 7, 2024

Published: April 17, 2024

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pgen.1011212

Copyright: © 2024 Timmins et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The summary statistics on prostate cancer are currently available via dbGaP at phs001120.v2.p2, or via application to PRACTICAL (contact: PRACTICAL@icr.ac.uk).

RESEARCH ARTICLE

Bayesian approach to assessing population differences in genetic risk of disease with application to prostate cancer

Iain R. Timmins^{1,2,3}*, The PRACTICAL Consortium¹, Frank Dudbridge¹

1 Department of Population Health Sciences, University of Leicester, Leicester, United Kingdom, 2 Division of Genetics and Epidemiology, The Institute of Cancer Research, London, United Kingdom, 3 Statistical Innovation, AstraZeneca, Cambridge, United Kingdom

¶ Membership of the PRACTICAL consortium is provided in Supporting Information file S1 Acknowledgements.

* iain.timmins@btinternet.com

Abstract

Population differences in risk of disease are common, but the potential genetic basis for these differences is not well understood. A standard approach is to compare genetic risk across populations by testing for mean differences in polygenic scores, but existing studies that use this approach do not account for statistical noise in effect estimates (i.e., the GWAS betas) that arise due to the finite sample size of GWAS training data. Here, we show using Bayesian polygenic score methods that the level of uncertainty in estimates of genetic risk differences across populations is highly dependent on the GWAS training sample size, the polygenicity (number of causal variants), and genetic distance (F_{ST}) between the populations considered. We derive a Wald test for formally assessing the difference in genetic risk across populations, which we show to have calibrated type 1 error rates under a simplified assumption that all SNPs are independent, which we achieve in practise using linkage disequilibrium (LD) pruning. We further provide closed-form expressions for assessing the uncertainty in estimates of relative genetic risk across populations under the special case of an infinitesimal genetic architecture. We suggest that for many complex traits and diseases, particularly those with more polygenic architectures, current GWAS sample sizes are insufficient to detect moderate differences in genetic risk across populations, though more substantial differences in relative genetic risk (relative risk > 1.5) can be detected. We show that conventional approaches that do not account for sampling error from the training sample, such as using a simple t-test, have very high type 1 error rates. When applying our approach to prostate cancer, we demonstrate a higher genetic risk in African Ancestry men, with lower risk in men of European followed by East Asian ancestry.

Author summary

Many diseases and complex traits, such as prostate cancer, exhibit differences in incidence across populations. Yet the potential contribution of genetic factors towards such

Funding: FD is funded by the MRC (MR/S037055/ 1, https://mrc.ukri.org/). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: IRT is a full-time employee of AstraZeneca.

disparities is unclear. Polygenic scores summarise genetic effects across the genome and can in principle provide a valuable tool for assessing and comparing disease risk across populations. In practise, current approaches based on polygenic scores assume that such scores perfectly measure genetic risk of disease without measurement error, and thus do not account for uncertainty that arises in the construction of the score from a finite genome-wide association study (GWAS) training sample, which can be substantial. We introduce a Bayesian approach based on the LDpred2 polygenic score model that accounts fully for training sample uncertainty, and we propose a Wald test for formally testing such genetic risk differences across populations. Simulations show that the method properly controls for type 1 errors assuming independent SNPs (achieved by pruning), and that statistical power is sensitive to both the genetic architecture (heritability and polygenicity) and training sample size. In application to prostate cancer, this framework enables us to identify a higher genetic risk in African Ancestry men, with lower risk in men of European followed by East Asian ancestry.

Introduction

An important open question in genetics is the degree to which population differences in disease risk can be attributed to differences in genetic risk. This question has recently been considered for several diseases and complex traits [1-7] including prostate cancer [8].

One commonly performed approach uses *t*-tests and ANOVA to examine differences in mean polygenic scores. Using this method, Conti et al. [8] estimated a 2.18-times higher genetic risk for prostate cancer in African men in comparison with European ancestry men, and a 0.73-times lower genetic risk in East Asian compared to European ancestry men. Similarly, Morris et al. [9] found evidence of a higher genetic risk of lupus in East Asian compared to most other global populations. In each case, using polygenic scores, the authors have drawn strong conclusions about differences in disease risk being attributable to different underlying genetic risk profiles.

However, we argue that these approaches are problematic, because they do not account for uncertainty that arises in the construction of the polygenic score itself. In effect, using *t*-tests and ANOVA approaches in this manner assumes that polygenic scores measure genetic risk of disease without any measurement error. This would be appropriate for estimating the accuracy of a given polygenic score between populations, such as to assess prospects for risk prediction. However for inference about the underlying genetic risk, we must allow that polygenic scores are derived from a finite sample of GWAS training data, and as shown previously [10–12], genetic risk estimates based on polygenic scores carry large amounts of uncertainty which should be accounted for in subsequent analyses.

To better understand and explain this source of uncertainty from the GWAS training data, consider a scenario where we repeatedly perform a GWAS analysis, and then create polygenic scores and calculate their mean difference between two population samples. Then across such repeats there will be variation in the estimates of genetic risk difference. This variation will, among other factors, be dependent upon the initial GWAS sample size. Since previous studies do not account for such inferential variation from the GWAS training data, one can question the certainty of conclusions drawn.

To address this issue, we develop a framework for estimating the genetic contribution to the relative risk of disease across populations that accounts for uncertainty in the construction of the polygenic score. We examine how the uncertainty in estimates of relative genetic risk across populations depends on the genetic architecture of the disease, the training sample size, and the genetic distance (F_{ST}) between populations compared. We further derive a Wald test statistic for formally testing differences in genetic risk across populations, which we show to have good control of type 1 errors under a simplified assumption of independent SNPs, which in practise we achieve through a pruning approach. We stress that future research should seek to extend this method further to account for the realistic effects of linkage disequilibrium (LD) and imperfect tagging. Finally, we also re-analyse GWAS summary data from Conti et al. [8] on prostate cancer using a genome-wide polygenic score derived using LDpred2 [13], ensuring that uncertainty in the training sample is taken into account.

Description of the Method

Bayesian approach to estimating relative risk for populations

We use a Bayesian approach to estimating relative genetic risk across populations. Let y_i be a trait measured on the *i*th individual, x_i an $M \times 1$ vector of genotypes (equal to 0, 1 or 2) and $\boldsymbol{\beta}$ an $M \times 1$ vector of per-allele effects for each genetic variant. We consider a linear model $y_i = \mathbf{x}_i^T \boldsymbol{\beta} + \epsilon_i$. The genetic effects are estimated as $\hat{\boldsymbol{\beta}}_{GWAS}$, denoting the marginal estimates from GWAS summary statistics. We use the Bayesian approach of LDpred2 [13], which assumes the effects at SNP *j* are drawn from a mixture distribution, where f_j denotes the allele frequency in the training sample:

$$\beta_j \sim \begin{cases} N \left(0, \frac{h_g^2}{M p_{\text{causal}}[2f_j(1-f_j)]} \right), & \text{with probability } p_{\text{causal}} \\ 0, & \text{with probability } 1 - p_{\text{causal}} \end{cases}$$

By combining the prior distribution $p(\boldsymbol{\beta}|h_g^2, p_{causal})$ and the likelihood of the observed data $p(\hat{\boldsymbol{\beta}}_{GWAS}|\boldsymbol{\beta})$, we can compute a posterior distribution as $p(\boldsymbol{\beta}|\hat{\boldsymbol{\beta}}_{GWAS}; h_g^2, p_{causal})$. We use $\tilde{\boldsymbol{\beta}} \sim p(\boldsymbol{\beta}|\hat{\boldsymbol{\beta}}_{GWAS}; h_g^2, p_{causal})$ to refer to the samples from the posterior distribution. The genetic value for individual *i* is estimated through the Bayesian polygenic score, defined as $P\hat{GS}_i = \boldsymbol{x}_i^T \mathbb{E}[\boldsymbol{\beta}|\hat{\boldsymbol{\beta}}_{GWAS}; h_g^2, p_{causal}]$.

Suppose further that we wish to find the relative genetic risk between two populations. We assume that the genetic effects are the same in both populations, so that differences in genetic risk arise only from differences in allele frequencies. Let the allele frequencies for variant j be f_j and g_j in populations 1 and 2. Using the result derived by Conti et al. [8], the relative risk comparing population 1 versus population 2 is $RR = \exp[d]$, where the difference d in mean genetic risk (on the log scale) across the populations is given by:

$$d = \mathbb{E}[\boldsymbol{x}_1^T \boldsymbol{\beta}] - \mathbb{E}[\boldsymbol{x}_2^T \boldsymbol{\beta}] = \sum_{j=1}^M 2(f_j - g_j)\beta_j$$

Next, we demonstrate that the posterior variance, var $[d|\hat{\boldsymbol{\beta}}_{GWAS}; h_g^2, p_{causal}]$ can be evaluated analytically under a simplified assumption where all SNPs are assumed independent. Under the non-infinitesimal model, we have derived analytical expressions for the posterior variance of individual SNP effect sizes var $[\beta_i|\hat{\beta}_{GWAS,i}, h_g^2, p_{causal}]$ (see S1 Appendix). Using these, and under the assumption of independent SNPs, the posterior variance of d can be expressed as:

$$\operatorname{var}[d|\hat{\boldsymbol{\beta}}_{\mathrm{GWAS}}; h_{g}^{2}, \boldsymbol{p}_{\mathrm{causal}}] = \sum_{1}^{M} 4(f_{j} - g_{j})^{2} \operatorname{var}[\beta_{j}|\hat{\beta}_{\mathrm{GWAS},j}; h_{g}^{2}, \boldsymbol{p}_{\mathrm{causal}}]$$
(1)

We demonstrate the validity of our analytical expressions for the posterior variance of *d* through comparisons with LDpred2-auto for a range of genetic architectures and GWAS training sample sizes (S1 Fig).

For the case of an infinitesimal model, we show further that the analytical form can be approximated by a simple closed-form expression (see S2 Appendix):

$$\operatorname{var}[d|\hat{\boldsymbol{\beta}}_{\mathrm{GWAS}}, h_g^2] \approx \frac{4MF_{ST}}{N} \left(1 + \frac{M}{Nh_g^2}\right)^{-1}$$
(2)

We demonstrated the validity of the closed-form analytical expressions for the posterior variance of *d* through comparisons with LDpred2-grid (with p_{causal} set to 100%), showing the formula has the correct functional dependence with each of the parameters h_g^2 , M, N and F_{ST} (S2 Fig).

Additionally, to test for differences in genetic risk across populations, we consider a Wald test on the posterior mean estimator $\hat{d} = \mathbb{E}[d|\hat{\boldsymbol{\beta}}_{\text{GWAS}}; h_{g}^{2}, p_{\text{causal}}]$ (S3 Appendix).

Verification and comparison

Simulation study

We simulated a range of genetic architectures and training sample sizes to understand the expected levels of uncertainty in estimates of *d*. To ensure our simulations were tractable at large samples, we simulated GWAS summary statistics, rather than individual-level phenotypes and genotypes.

Summary statistics were simulated using a wide range of values of SNP-heritability h_g^2 , polygenicity p_{causal} genetic differentiation F_{ST} , and effective training sample size N_{eff} . We assumed the genetic architecture could be described by a set of M = 200,000 SNPs, which were independent (i.e., in linkage equilibrium) and directly genotyped in both populations. This best-case scenario gives a lower bound on the uncertainty expected in real data applications.

We generated allele frequencies using a version of the Balding-Nichols model [14] where we assume that the ancestral allele frequency is unknown [15]. We assume a simplified model where all SNPs are independent. For each SNP, the allele frequency f_j in the first population was drawn from the uniform distribution on [0.1,0.9], and the allele frequency g_j of the second populations was drawn from a beta distribution with parameters $f_j(1-2F_{ST})/2F_{ST}$ and $(1-f_j)(1-2F_{ST})/2F_{ST}$.

We considered a range of genetic architectures: a heritability h_g^2 of 0.05, 0.10, 0.25, 0.50, 0.80, polygenicity p_{causal} of 0.1%, 1%, 10%, and 100%, genetic differentiation F_{ST} of 0.02, 0.04, 0.06, 0.08, 0.10 and 0.12, and we varied the effective sample size N_{eff} between 10³ and 10⁸.

For each simulation replicate, SNP effect sizes were drawn from the non-infinitesimal mixture model, based on the allele frequencies of population 1:

$$\beta_j \sim \begin{cases} N \bigg(0, \frac{h_g^2}{M p_{\text{causal}}[2f_j(1-f_j)]} \bigg), & \text{with probability } p_{\text{causal}} \\ 0, \text{with probability } 1 - p_{\text{causal}} \end{cases}$$

and marginal effect estimates were drawn from [16]:

$$\hat{\boldsymbol{\beta}}_{\text{GWAS},j} | \boldsymbol{\beta}_j \sim N\left(\boldsymbol{\beta}_j, \frac{1}{2Nf_j(1-f_j)}\right)$$

where it was assumed that the effect sizes are estimated based on samples predominantly taken from population 1. We used the derived analytical expressions to efficiently evaluate the posterior mean and variance of *d* without recourse to MCMC simulations, where we set p_{causal} and h_g^2 equal to their true values. For each set of parameters, we estimated the expected posterior variance of *d* by averaging results across 100 summary statistic replicates. The steps for performing this simulation study are provided in S5 Appendix.

Additionally, for each set of parameters, we evaluated the expected sampling variance, bias and mean square error of the posterior mean \hat{d} , also by averaging across 100 simulations. The steps of this simulation study evaluating the properties of \hat{d} are provided in <u>S6 Appendix</u>.

To understand the type 1 error for our Wald test on d, we generated χ^2 -test statistics for each simulation under the null hypothesis d = 0. We also evaluated the statistical power of the Wald test, where we simulated from scenarios corresponding to a relative risk ($RR = \exp[d]$) of 1.1, 1.2, 1.5 and 2, whilst varying both the training sample size and genetic architecture. We also evaluated the type 1 error of the *t*-test, where we varied the target sample size between 5,000, 10,000, 20,000, and 100,000 for both of the two target populations. For statistical power and type 1 error calculations, we used 1,000 simulation replicates. The steps of this simulation study for evaluating the type 1 error and power of the Wald test are also provided in S6 Appendix.

Results

The main expressions derived above are the Bayesian posterior variance of the difference in polygenic scores between populations, and in particular, the closed-form expression for this variance under an infinitesimal model. Here we explore the properties of the posterior variance and use simulations to confirm the derived expressions.

Posterior variance of genetic relative risk

We illustrate the impact of the training sample size and genetic architecture on estimating the genetic relative risk ($RR = \exp[d]$) between populations using a range of summary statistic simulations. The uncertainty in estimates of genetic relative risk across populations, as characterised by the posterior standard deviation s.d.(d), is presented in Fig 1.

First, we fixed the training sample size, heritability, and genetic distance (F_{ST}) and varied the levels of polygenicity. We found that the uncertainty in estimates of genetic relative risk across populations increased sharply with the level of polygenicity. When we simulated summary statistics for a scenario where N = 100,000, $h_g^2 = 0.50$ and $F_{ST} = 0.10$ ($F_{ST} = 0.10$ is typical differentiation between divergent continental populations, such as European and African populations), the uncertainty s.d.(d) ranged from 0.04, 0.15, 0.35 to 0.40 when the proportion of causal variants varied from 0.1%, 1%, 10% to 100%, respectively.

Additionally, the level of uncertainty in estimates of genetic relative risk under an infinitesimal genetic architecture ($p_{causal} = 100\%$) was relatively high, even at very large sample sizes. Simulating summary statistic data with $p_{causal} = 100\%$, $h_g^2 = 0.50$ and $F_{ST} = 0.10$, we found that the uncertainty s.d.(*d*) decreased from 0.37 to 0.24, as the training sample size increased from 200,000 to 1,000,000.



Fig 1. Estimates of posterior s.d.(*d*) based on simulations. Heritability is held constant at $h_g^2 = 0.50$.

The level of uncertainty s.d.(*d*) also increased with genetic distance F_{ST} . For example, in the scenario where N = 100,000, $h_g^2 = 0.50$ and $p_{causal} = 10\%$, when we varied F_{ST} between 0.02, 0.06 and 0.10, the corresponding uncertainty s.d.(*d*) was 0.18, 0.31 and 0.40, respectively.

Closed-form approximation under infinitesimal model

As shown in Eq (2) above, we further derived a closed-form analytical estimate of the variance of $d = \log(RR)$ under an infinitesimal genetic architecture, where it is assumed that every SNP has effect sizes drawn from $\beta_j \sim N(0, h_g^2/[2Mf_j(1 - f_j)])$ (see methods and S2 and S3 Appendices). We examined the accuracy of this closed-form estimate where the true model was in fact non-infinitesimal, varying both the heritability and polygenicity (Fig 2).



Fig 2. Closed-form analytical estimator of posterior s.d.(*d*) from infinitesimal model compared with average posterior s.d.(*d*) from summary statistic simulations. The x-axis is the average posterior s.d.(*d*) from summary statistic simulations. The y-axis is the expected posterior s.d.(*d*) computed with Eq (2). Each dot represents the average of 100 simulations replicated for each p_{causal} . The number of causal variants was fixed at M = 200,000, the genetic differentiation F_{ST} was fixed at 0.10, the GWAS training sample size N was fixed at 100,000.

https://doi.org/10.1371/journal.pgen.1011212.g002

We found the closed-form estimates were broadly comparable with the expected variance for levels of true polygenicity between 10% and 100%, although there were clear overestimates of the posterior variance when the true polygenicity was below 10%. For instance, for a heritability of $h_g^2 = 0.50$, genetic distance $F_{ST} = 0.10$, and training sample size N = 100,000, the closed-form approximation gave s.d.(d) = 0.40. Meanwhile, when we simulated summary statistics and used the analytic form of the posterior variance per-SNP (see S1 Appendix), we estimated s.d.(d) as 0.04, 0.15, 0.35 and 0.40 as we varied the polygenicity between 0.1%, 1%, 10% and 100%, respectively.

This suggested that for genetic architectures with low polygenicity, we should instead use analytical estimates of the posterior variance per-SNP (see S1 Appendix) or MCMC simulations to derive the posterior distribution of d, rather than the closed-form approximation of Eq (2).

Assessment of type 1 error, statistical power and bias

Additionally, we assessed the frequentist properties of the Bayesian posterior mean estimator \hat{d} , with results presented for a range of scenarios in Figs 3 and 4.

First, having derived a Wald test statistic that accounts for uncertainty relating to the training sample, we demonstrated that this test is well-calibrated, having the expected type 1 error rates (Fig 3). In comparison to our Wald test, we noted that the *t*-test has very high type 1 error rates unless the training sample size is very large. For example, performing a *t*-test using a target sample size of N = 5,000 for both of the populations, and assuming a genetic architecture with $h_g^2 = 0.50$, $p_{causal} = 10\%$ and genetic distance $F_{ST} = 0.10$, we note that a training sample size in excess of 10^8 is needed to achieve a type 1 error of 0.05. We additionally observed that the *t*-test has the property of being particularly sensitive to scenarios where the target sample size is large relative to the training sample. In these cases, while the training sample may be large enough to provide a precise polygenic score, the *t*-test is still able to detect differences in population means that are due to residual sampling error from the construction of the polygenic score (relating to the training sample size), even when no true difference exists. This leads to an unintuitive finding that the type 1 error can be better controlled for the *t*-test at smaller target samples sizes. Nonetheless, the type 1 error rates for the *t*-test remain uniformly higher than the Wald test (Fig 3).

Second, we estimated the statistical power of the Wald test using simulations, with results for a range of typical scenarios presented in Fig 4A, where we fixed the heritability at $h_g^2 = 0.50$ and the genetic distance at $F_{ST} = 0.10$, and estimated the power for a true relative risk (exp [*d*]) between 1.1, 1.2, 1.5 and 2. We found a strong dependence between the statistical power and proportion of causal variants, p_{causal} . For a training sample size of N = 100,000, and assuming a true relative risk of RR = 1.5 ($d = \log (1.5)$), the estimated power of the Wald test varied between 1.00, 0.79, 0.18 and 0.07 as the polygenicity increased from 0.1%, 1%, 10% to 100%, respectively. Fixing the polygenicity at $p_{causal} = 10\%$ (typical of most complex trait genetic architectures), for a training sample size of N = 100,000, the power was estimated as 0.11, 0.13, 0.18 and 0.34 as we varied the true relative risk (exp [*d*]) between 1.1, 1.2, 1.5 to 2, respectively. Moreover, for the same polygenicity $p_{causal} = 10\%$ and a very large training sample size of N = 1,000,000, the power was estimated as 0.20, 0.40, 0.91 and 1.00 as we increased the true relative risk (exp [*d*]) between the values 1.1, 1.2, 1.5 and 2, respectively.

Third, we considered the bias and mean square error for the posterior mean estimator d, showing the bias-variance trade-off for a typical scenario in Fig 4B. For this example, the heritability was fixed at $h_g^2 = 0.50$ and the genetic distance at $F_{ST} = 0.10$, while in this case the true



Fig 3. Type 1 error rates for the Wald test and *t*-test. Simulation study estimates of the type 1 error rates for a Wald test and *t*-test of size $\alpha = 0.05$. Heritability is held constant at $h_g^2 = 0.50$, genetic distance F_{ST} fixed at 0.10. Polygenicity, and training and target sample sizes are varied as shown. The grey line represents the size $\alpha = 0.05$.

genetic relative risk was fixed at RR = 1.5 ($d = \log(1.5)$). We varied the training sample size and the proportion of causal variants p_{causal} . We note that since the posterior mean \hat{d} is a shrinkage estimator, at small training sample size the shrinkage is substantial and the bias very high. Hence, we observed in Fig 4B that the sampling variance of \hat{d} is low at small training sample sizes as the estimates of \hat{d} are drawn towards zero, whilst at the highest training sample size the sampling variance is proportional to 1/N, and between these extremes at intermediate sample sizes the combination of shrinkage and sampling error results in higher sampling variance. Additionally, the estimator is asymptotically unbiased, with the mean square error consistently converging to zero at large training sample sizes. Moreover, the rate that the bias converges to



Fig 4. Frequentist properties of the Bayesian posterior mean \hat{d} **for a range of scenarios.** (A) Statistical power for the Wald test with size $\alpha = 0.05$. Heritability is held constant at $h_g^2 = 0.50$, genetic distance F_{ST} fixed at 0.10. (B) Mean square error (MSE), bias (squared), and sampling variance of \hat{d} . Heritability is fixed at $h_g^2 = 0.50$, genetic distance F_{ST} at 0.10, and true genetic relative risk $RR = \exp[d]$ fixed at 1.5.

zero is slower for genetic architectures with higher polygenicity p_{causal} . For a typical genetic architecture with $p_{\text{causal}} = 10\%$, we observe visually that the bias remains high even at training samples of N = 100,000, falling rapidly at sample sizes of N = 1,000,000 and above.

Applications

Case study: Prostate cancer in PRACTICAL Consortium

We provide a real data example based on the analysis performed by Conti et al. [8], which estimated the difference in genetic risk of prostate cancer across European, Hispanic, South Asian, and African populations. We used summary statistics obtained from the multi-ancestry analysis which included 107,247 cases and 127,006 controls. We estimated the mean genetic risks based on allele frequencies from the control groups of each population within the PRAC-TICAL consortium. We used a set of 1,444,196 HapMap3+ SNPs [17], which we further filtered to a set of 234,822 approximately independent SNPs through clumping using the African (AFR, N = 504) population within the 1000 Genomes project [18] (1KGP), where we clumped based on allele frequency using the bigsnpr R package [19] snp_clumping command with $r^2 < 0.1$ and 1,000-kb windows.

As we did not have access to a test dataset for parameter tuning, we generated the PGS model using LDpred2-auto [13], also implemented using the bigsnpr R package [19]. This approach estimates hyperparameters from the training data, namely the proportion of causal variants p_{causab} and the SNP-heritability h_a^2 . We used LDpred2-auto to generate the polygenic score using a Gibbs sampling chain with 3,000 iterations after 1,000 burn-in, with SNP effect sizes averaged across all chains. We used 30 initial values for p_{causal} , ranging from 10^{-4} to 0.9. We computed the initial h_{α}^2 from the snp_ldsc function. We filtered chains by comparing the scale of the resulting predictions as described in the LDpred2 vignette (bigsnpr, version 1.12.2). Moreover, since the training sample was composed predominantly of European ancestry participants, we used the default EUR reference panel from UK Biobank, provided by Privé et al. [17]. Using the posterior samples $\tilde{\beta}^{(1)}, \tilde{\beta}^{(2)}, \ldots, \tilde{\beta}^{(3,000)}$ (returned using the report step = 1 argument in the snp_ldpred2_auto command), we generated posterior samples $\tilde{d}^{(1)}, \tilde{d}^{(2)}, \ldots$, $\tilde{d}^{(3,000)}$ comparing the mean difference in genetic values for European, Hispanic, South Asian and African ancestries, where the allele frequencies were based on the control group in each population. We evaluated the posterior mean and variance of d by averaging across these posterior samples. Finally, using our Wald test, we computed χ^2 -statistics and corresponding Pvalues for comparing the difference in genetic risk across each pair of populations. We also used our simulation approach (S6 Appendix) to evaluate the statistical power available for identifying genetic risk differences of a range of magnitudes (relative risk, RR varied between 1.1, 1.2, 1.5, 2), given the genetic architecture (h_{σ}^2 and p_{causal}), effective sample size N, and genetic distance F_{ST} for each population comparison.

The relative risk estimates across populations are shown in Table 1, where we compared results from our Wald test with those based on the *t*-test analysis performed previously [8]. The hyperparameters were estimated as $h_g^2 = 0.13$ and polygenicity $p_{causal} = 0.34\%$. In comparison to the mean genetic risk for men of European ancestry, we estimated that men of African ancestry had relative risks of 2.99 (95% credible interval, 1.32–6.78, $P = 2x10^{-3}$), while men of East Asian ancestry and Hispanic men had relative risks of 0.46 (95% CI, 0.18–1.19, P = 0.05) and 0.82 (95% CI, 0.60–1.12, P = 0.17), respectively. Hence, we found broadly similar point estimates to the original Conti et al. [8] analysis, with greater uncertainty that reflects accounting for the GWAS training sample size.

We further considered the statistical power available for analysing each of these contrasts using our Wald test, given the observed genetic architecture, effective sample size and genetic distance across populations (Table 1). Using our simulation approach (S6 Appendix), we fixed $h_g^2 = 0.13$, $p_{causal} = 0.34$, and the effective sample size at N = 232,586. For the comparison of European and African populations, using our derived estimator for F_{ST} (S4 Appendix) which

Population 1	Population 2	Relative genetic risk*				Statistical Power for Wald test				
		<i>t</i> -test, Conti <i>et al.</i> 2021 [8]		Wald test		$h_g^2 = 0.13, p_{\text{causal}} = 0.34\%, N = 232,586$				
		Relative risk (95% Confidence intervals)	P- value**	Relative risk (95% Credible intervals)	P- value	<i>FST</i> ***	<i>RR</i> = 1.1	<i>RR</i> = 1.2	<i>RR</i> = 1.5	<i>RR</i> = 2.0
African	European	2.18 (2.14-2.22)	$< 10^{-300}$	2.99 (1.32-6.78)	$2x10^{-3}$	0.12	0.46	0.84	1.00	1.00
East Asian	European	0.73 (0.71-0.76)	2x10 ⁻⁷³	0.46 (0.18–1.19)	0.05	0.12	0.46	0.84	1.00	1.00
Hispanic	European	0.97 (0.94-1.00)	0.03	0.82 (0.60–1.12)	0.17	0.02	0.94	1.00	1.00	1.00

Table 1. Comparison of prostate cancer genetic relative risk estimates across populations.

Relative risk estimates from Conti et al. 2021 [8] t-test with our Wald test.

* $RR = \exp[d]$, here $d = \mu_1 - \mu_2$, and μ_1 and μ_2 denote the polygenic score mean in population 1 and 2, respectively.

** P-value calculated based on normal approximation on the log-hazard scale using confidence intervals from Conti et al. 2021 [8].

*** F_{ST} was evaluated using our estimator derived in S4 Appendix.

https://doi.org/10.1371/journal.pgen.1011212.t001

gave $F_{ST} = 0.12$, we estimated the power to be 0.46, 0.84 and 1.00 for identifying a genetic relative risk of 1.1, 1.2 and 1.5, respectively, showing the test to be well-powered. Similar results were found for the comparison of European and East Asian populations, as the population distance was also estimated as $F_{ST} = 0.12$. Additionally, for the comparison of European and Hispanic populations, we found $F_{ST} = 0.02$, which led to estimates of power of 0.94 and 1.00 for identifying a genetic relative risk of 1.1 and 1.2, respectively, hence our Wald test for this contrast was also well-powered.

Case study: UK Biobank

We provide a further realistic application of our method to assess the impact of the pruning procedure on the accuracy of s.d.(*d*). We considered a set of 28 diseases and complex traits from UK Biobank, which we based approximately on sets of phenotypes compiled in previous studies [20,21], which were representative of a diverse range of genetic architectures. We used summary statistics from GWAS conducted from Neale Lab (nealelab.is/uk-biobank), which were derived from N = 361,194 individuals of white-British ancestry. For these phenotypes, we evaluated s.d.(*d*) using the pruned set of 234,822 SNPs defined above ($r^2 < 0.1$ and 1,000-kb), where we compared the risk differences across European and African populations. We computed s.d.(*d*) using LDpred2-auto with 3,000 iterations after 1,000 burn-in. For the hyperparameters p_{causal} and h_g^2 estimated using LDpred2-auto, we further computed s.d.(*d*) using Eq (1), which assumes SNPs are independent. This enabled us to assess how sensitive our estimates of s.d.(*d*) were to any residual correlation that remains across SNPs after the pruning process.

The results for this sensitivity analysis in UK Biobank are presented in Fig 5 and S1 Table. As expected we observed that using Eq (1), which assumes independence between SNPs, there were slightly higher estimates of posterior standard deviation s.d.(d) than those obtained from the LDpred2-auto approach that accounts for correlation between the SNPs. In particular, using Eq (1), the estimates of s.d.(d) across 28 phenotypes increased on average by 13.5% in comparison to LDpred2-auto. This shows that our analytical formula leads to slight overestimates of s.d.(d). Given this finding, our earlier simulation study results (see Verification and Comparison) based on sets of independent rather than pruned SNPs will provide slight overestimates of the true s.d.(d), and are hence mildly conservative regarding the statistical power and type 1 error rates of the Wald test that we would expect in realistic settings.



Fig 5. Validation of posterior variance formula in UK Biobank. Estimates of the posterior standard deviation s.d.(*d*) comparing European and African population disease risk for 28 diseases and complex traits from UK Biobank. For the x-axis, s.d.(*d*) is computed under Eq (1), which assumes the SNPs are independent, and for the y-axis, s.d.(*d*) is computed using LDpred2-auto which accounts for the correlation between the pruned SNPs.

Discussion

In this study we develop a framework for understanding the genetic contribution to differences in disease risk across populations. Unlike previous studies, we demonstrate the need to account for uncertainty relating to the size of the GWAS training sample. We further show that uncertainty in estimates of genetic risk differences across populations depends heavily upon the genetic architecture of the phenotype, which is comprised of both the heritability and proportion of causal variants, as well as being dependent upon the genetic distance (F_{ST}) between the two populations considered. Moreover, we used a Bayesian framework for estimating relative genetic risk across populations, as Bayesian approaches deliver the highest accuracy for genetic risk prediction [22,23], and are highly flexible in incorporating information about the genetic architecture [24,25].

Using a range of simulation studies, we found a strong relationship between the polygenicity of the genetic architectures and the uncertainty in estimates of genetic relative risk across populations. For phenotypes with polygenicity in the typical range for most diseases and complex traits (between 1% and 10% of variants causal), the uncertainty was low only at sample sizes >200,000. Hence for most diseases and complex traits, we argue that the sample size present in current biobank-scale GWAS studies and large meta-analyses is too small to clearly discriminate small genetic risk differences between global populations (i.e., relative risk between 1 and 1.5).

For the case of the infinitesimal model, where all variants are assumed to have a causal effect on the phenotype, we further derive a closed-form expression that demonstrates clearly how uncertainty in estimates of relative genetic risk is a function of the heritability, training data sample size, and genetic distance (F_{ST}) between populations. We showed that for non-infinitesimal models with a high proportion of causal variants, this simple closed-form expression provided a reasonable estimate of the posterior variance. However, we still recommend the use of either analytical results for the posterior distribution, which we derived, or MCMC simulations to estimate posterior variances, as these approaches provide greater accuracy and are straightforward to implement [13].

We further derived a Wald test statistic relating to the posterior mean estimator of the log relative risk across populations. We demonstrated that this test statistic had well-calibrated type 1 error rates based on simulations, while the power of the test depended strongly upon the proportion of causal variants along with the training sample size. For more polygenic complex trait genetic architectures ($p_{causal} \sim 10\%$), we observed that there was limited statistical power available at current GWAS sample sizes (~100,000) to detect moderate differences in genetic relative risk across populations (relative risk of 1.5 to 2), while to detect smaller differences in risk (relative risk of 1.1 to 1.2) the power only became sufficiently large for sample size more than 1,000,000. In contrast, for less polygenic traits and diseases ($p_{causal} \sim 1\%$), there was in general sufficient power available at current available sample sizes to detect more modest differences in genetic risk across populations, as we also showed in our prostate cancer example. We note further that the conventional *t*-test, which compares mean genetic risk differences by evaluating polygenic scores in separate target samples for each population, has very poorly controlled type 1 error rates, even when the initial training sample size is very large. This is due to the *t*-test detecting mean differences in scores that are due to sampling error in the construction of the polygenic score from the training sample, even when there are no underlying true differences in risk across the populations. This effect becomes more severe as the target sample size increases. This is clearly a problematic property of the *t*-test, and this work provides strong evidence against using this approach.

Having developed a framework for understanding the uncertainty in estimates of relative genetic risk for a general non-infinitesimal model, we provide a real data example by re-analysing the study by Conti et al. [8] on prostate cancer risk. The original analysis by the authors involved computing polygenic scores for prostate cancer risk across samples of participants from several populations, and then using *t*-tests to compare the relative risk across pairs of populations. Such an approach assumes the polygenic scores, which were composed of only 269 variants, capture the complete genetic risk of prostate cancer for individuals without measurement error, which is not a plausible assumption. However, our approach continued to identify a significantly higher genetic risk for males of African compared to European ancestry, and a lower risk for males of East Asian ancestry. Our findings were broadly comparable with the relative risks derived in the original study, with much greater uncertainty surrounding the point estimates, as expected, since uncertainty in the SNP effect estimates arising from the finite GWAS training sample was fully accounted for.

Our study has limitations to note. First, there are several challenges that exist in interpreting differences in polygenic score means across populations. Observed differences could be attributable to various forms of bias, such as uncorrected population stratification [26-28], as well as the choice of discovery data and polygenic score method [29]. While we were unable to

consider the sensitivity of our test statistic to these forms of bias, our best-case scenario provides an informative lower bound on the levels of uncertainty in estimates of population differences in genetic risk.

Second, our approach depends upon using an independent set of variants to model genetic risk differences, avoiding the complexities of modelling linkage disequilibrium patterns across populations. This however can exacerbate the issue of imperfect tagging of causal variants, as information is lost during the pruning process, dampening the predictive performance of the polygenic score. The magnitude of this effect is difficult to ascertain, although we note that LDpred2 estimates have been shown to be robust to the impact of imperfect tagging [12], while only modest reductions in predictive accuracy are lost when pruning markers for linkage disequilibrium [30]. Differences in genetic risk identified by our approach may simply reflect differences in LD with unknown causal variants. We did however try to mitigate the effect of LD differences in the prostate cancer data by using an African (rather than European) ancestry reference panel for the pruning procedure to maximise coverage of causal variants. Nonetheless, we must base our inferences on a set of largely tagged SNPs, and we must further assume that any such allele frequency differences on the tagged SNPs will reflect the frequency differences at the true underlying causal SNPs. Future work should aim to perform simulations and case studies with realistic LD patterns that assess how estimates of population genetic risk differences taken from a set of imperfectly tagged markers, such as the effect estimate on a set of pruned SNPs, compares with the true population difference based on the complete set of causal SNPs.

Third, we have assumed that the causal effect sizes are equal across populations, and hence we focus on differences in disease risk that are attributable to allele frequency differences (i.e., caused by random drift or selection). This is consistent with findings from recent studies which have shown that, when environments are well controlled, there is minimal heterogeneity in causal effects by ancestry [31,32]. However, our approach lacks the flexibility to account for diseases or complex traits that have population-specific genetic architectures, since differences in genetic risk. In these cases, incorporating parameters such as a cross-population genetic correlation [38,39] could be one potential approach, although this would require summary statistics with large sample sizes for at least two populations, which are not always available. Note however that, in our model, it is the product of genotype and effect size that contributes to genetic risk. Thus, if causal effect sizes are truly different, the genetic risk is the same as it would be from variants with the same effect sizes but different allele frequencies. The result of different effect sizes, and also of imperfect tagging, is to increase the effective F_{ST} in our model, and we can thus accommodate these factors to some extent.

Additionally, we have focussed on variability in the training sample, whereas previous studies have only focussed on variability in the target sample. The latter approach is appropriate for comparing a given polygenic score between populations, such as for risk prediction. But for inference about the underlying genetic model it is necessary to allow also for training sample variance. A fuller account would allow for both training and target sample variance. We have not pursued this here as the extension of the Bayesian estimation to a target sample is not trivial; however an ad hoc Wald test could be constructed by summing the sampling variances of the training and target samples. In our prostate cancer analysis the results were essentially unchanged as the variance from the training sample was much greater than from the target sample.

Finally, while our approach can be used to detect true differences due to drift and selection, the impact of other potential sources of population differences, such as non-additive effects [40] and gene-environment interactions [41], and confounding by genetic and environmental effects that correlate with ancestry [42], is unclear.

In summary, we have developed a framework for understanding the genetic contribution to differences in disease risk across populations. We show that uncertainty in estimates of genetic relative risk across populations is highly dependent on the genetic architecture of the disease, particularly the polygenicity, as well as the training GWAS sample size. Accounting for the training sample size, we find evidence that population differences in prostate cancer risk are at least partly attributable to genetic factors.

Supporting information

S1 Appendix. Analytical form of posterior mean and variance of SNP effect sizes under non-infinitesimal model.

(DOCX)

S2 Appendix. Closed-form expression for posterior variance under infinitesimal model. (DOCX)

S3 Appendix. Wald test on the posterior mean estimator $\hat{d} = \mathbb{E}[d|\hat{\beta}_{GWAS}; h_g^2, p_{causal}]$. (DOCX)

S4 Appendix. Estimation of *F*_{*ST*}. (DOCX)

S5 Appendix. Simulation study steps for estimating the expected var $[d|\hat{\beta}_{GWAS;}, h_g^2, p_{causal}]$. (DOCX)

S6 Appendix. Simulation study steps for estimating the bias, sampling variance, and mean square error of \hat{d} , and the type 1 and 2 error for the Wald test and *t*-test on \hat{d} . (DOCX)

S1 Table. Results from case study in UK Biobank of posterior variance formula. (DOCX)

S1 Fig. Validation of posterior variance formula for *d* using LDpred2-auto MCMC simulations. Posterior standard deviation from analytical expressions (Methods and S1 Appendix) compared with estimates from LDpred2-auto using 1,000 MCMC samples after 100 burn-in iterations. F_{ST} was fixed at 0.10. Training sample size was varied between 10^4 and 10^8 . For each set of parameters, the expected posterior s.d.(*d*) was estimated by averaging across 100 simulations.

(TIFF)

S2 Fig. Validation of closed-form expression for posterior variance under infinitesimal model. Posterior standard deviation from closed-form expressions for infinitesimal model (S2 Appendix) compared with estimates from LDpred2-grid (with $p_{causal} = 100\%$) using 1,000 MCMC samples after 100 burn-in iterations. In panels A, B, C and D, each of the parameters N, h_g^2 , M and F_{ST} , respectively, are varied while the others are held fixed, with a few scenarios shown in each case. The closed-form expression is plotted as the curved line:

$$\operatorname{var}[d] = \frac{4MF_{ST}}{N} \left(1 + \frac{M}{Nh_g^2}\right)^{-1}$$
(TIF)

S1 Spreadsheet. Quantification of data shown in Fig 1. (XLSX)

S2 Spreadsheet. Quantification of data shown in Fig 2. (XLSX)

S3 Spreadsheet. Quantification of data shown in Fig 3. (XLSX)

S4 Spreadsheet. Quantification of data shown in Fig 4. (XLSX)

S5 Spreadsheet. Quantification of data shown in Fig 5. (XLSX)

S6 Spreadsheet. Quantification of data shown in <u>S1 Fig</u>. (XLSX)

S7 Spreadsheet. Quantification of data shown in <u>S2 Fig</u>. (XLSX)

S1 Acknowledgements. Members of the PRACTICAL consortium. (DOCX)

Author Contributions

Conceptualization: Iain R. Timmins, Frank Dudbridge.

Data curation: Iain R. Timmins.

Formal analysis: Iain R. Timmins.

Funding acquisition: Frank Dudbridge.

Investigation: Frank Dudbridge.

Methodology: Iain R. Timmins, Frank Dudbridge.

Software: Iain R. Timmins.

Supervision: Frank Dudbridge.

Writing - original draft: Iain R. Timmins.

Writing - review & editing: Iain R. Timmins, Frank Dudbridge.

References

- Turchin MC, Chiang CW, Palmer CD, Sankararaman S, Reich D, Genetic Investigation of ATC, et al. Evidence of widespread selection on standing variation in Europe at height-associated SNPs. Nat Genet. 2012; 44(9):1015–9. Epub 2012/08/21. https://doi.org/10.1038/ng.2368 PMID: 22902787; PubMed Central PMCID: PMC3480734.
- Berg JJ, Coop G. A population genetic signal of polygenic adaptation. PLoS Genet. 2014; 10(8): e1004412. Epub 2014/08/08. https://doi.org/10.1371/journal.pgen.1004412 PMID: 25102153; PubMed Central PMCID: PMC4125079.
- Robinson MR, Hemani G, Medina-Gomez C, Mezzavilla M, Esko T, Shakhbazov K, et al. Population genetic differentiation of height and body mass index across Europe. Nat Genet. 2015; 47(11):1357– 62. Epub 2015/09/15. https://doi.org/10.1038/ng.3401 PMID: 26366552; PubMed Central PMCID: PMC4984852.
- Mao L, Fang Y, Campbell M, Southerland WM. Population differentiation in allele frequencies of obesity-associated SNPs. BMC Genomics. 2017; 18(1):861. Epub 2017/11/12. https://doi.org/10.1186/ s12864-017-4262-9 PMID: 29126384; PubMed Central PMCID: PMC5681842.

- Guo J, Wu Y, Zhu Z, Zheng Z, Trzaskowski M, Zeng J, et al. Global genetic differentiation of complex traits shaped by natural selection in humans. Nat Commun. 2018; 9(1):1865. Epub 2018/05/16. https:// doi.org/10.1038/s41467-018-04191-y PMID: 29760457; PubMed Central PMCID: PMC5951811.
- Martin AR, Gignoux CR, Walters RK, Wojcik GL, Neale BM, Gravel S, et al. Human Demographic History Impacts Genetic Risk Prediction across Diverse Populations. Am J Hum Genet. 2017; 100(4):635–49. Epub 2017/04/04. https://doi.org/10.1016/j.ajhg.2017.03.004 PMID: 28366442; PubMed Central PMCID: PMC5384097.
- Curtis D. Polygenic risk score for schizophrenia is more strongly associated with ancestry than with schizophrenia. Psychiatr Genet. 2018; 28(5):85–9. Epub 2018/08/31. <u>https://doi.org/10.1097/YPG.</u> 00000000000206 PMID: 30160659.
- Conti DV, Darst BF, Moss LC, Saunders EJ, Sheng X, Chou A, et al. Trans-ancestry genome-wide association meta-analysis of prostate cancer identifies new susceptibility loci and informs genetic risk prediction. Nat Genet. 2021; 53(1):65–75. Epub 2021/01/06. https://doi.org/10.1038/s41588-020-00748-0 PMID: 33398198; PubMed Central PMCID: PMC8148035.
- Morris DL, Sheng Y, Zhang Y, Wang YF, Zhu Z, Tombleson P, et al. Genome-wide association metaanalysis in Chinese and European individuals identifies ten new loci associated with systemic lupus erythematosus. Nat Genet. 2016; 48(8):940–6. Epub 2016/07/12. https://doi.org/10.1038/ng.3603 PMID: 27399966; PubMed Central PMCID: PMC4966635.
- Dudbridge F. Power and predictive accuracy of polygenic risk scores. PLoS Genet. 2013; 9(3): e1003348. Epub 2013/04/05. https://doi.org/10.1371/journal.pgen.1003348 PMID: 23555274; PubMed Central PMCID: PMC3605113.
- Chatterjee N, Wheeler B, Sampson J, Hartge P, Chanock SJ, Park JH. Projecting the performance of risk prediction based on polygenic analyses of genome-wide association studies. Nat Genet. 2013; 45 (4):400–5, 5e1-3. Epub 2013/03/05. https://doi.org/10.1038/ng.2579 PMID: 23455638; PubMed Central PMCID: PMC3729116.
- Ding Y, Hou K, Burch KS, Lapinska S, Prive F, Vilhjalmsson B, et al. Large uncertainty in individual polygenic risk score estimation impacts PRS-based risk stratification. Nat Genet. 2022; 54(1):30–9. Epub 2021/12/22. https://doi.org/10.1038/s41588-021-00961-5 PMID: <u>34931067</u>; PubMed Central PMCID: PMC8758557.
- Prive F, Arbel J, Vilhjalmsson BJ. LDpred2: better, faster, stronger. Bioinformatics. 2020. Epub 2020/ 12/17. https://doi.org/10.1093/bioinformatics/btaa1029 PMID: 33326037; PubMed Central PMCID: PMC8016455.
- Balding DJ, Nichols RA. A method for quantifying differentiation between populations at multi-allelic loci and its implications for investigating identity and paternity. Genetica. 1995; 96(1–2):3–12. Epub 1995/ 01/01. https://doi.org/10.1007/BF01441146 PMID: 7607457.
- Bhatia G, Patterson N, Sankararaman S, Price AL. Estimating and interpreting FST: the impact of rare variants. Genome Res. 2013; 23(9):1514–21. Epub 20130716. <u>https://doi.org/10.1101/gr.154831.113</u> PMID: 23861382; PubMed Central PMCID: PMC3759727.
- Daetwyler HD, Villanueva B, Woolliams JA. Accuracy of predicting the genetic risk of disease using a genome-wide approach. PLoS One. 2008; 3(10):e3395. Epub 2008/10/15. https://doi.org/10.1371/ journal.pone.0003395 PMID: 18852893; PubMed Central PMCID: PMC2561058.
- Privé F, Albiňana C, Arbel J, Pasaniuc B, Vilhjálmsson BJ. Inferring disease architecture and predictive ability with LDpred2-auto. bioRxiv. 2023:2022.10.10.511629. <u>https://doi.org/10.1016/j.ajhg.2023.10.</u> 010 PMID: 37944514
- Genomes Project C, Auton A, Brooks LD, Durbin RM, Garrison EP, Kang HM, et al. A global reference for human genetic variation. Nature. 2015; 526(7571):68–74. Epub 2015/10/04. <u>https://doi.org/10.1038/nature15393</u> PMID: 26432245; PubMed Central PMCID: PMC4750478.
- Prive F, Aschard H, Ziyatdinov A, Blum MGB. Efficient analysis of large-scale genome-wide data with two R packages: bigstatsr and bigsnpr. Bioinformatics. 2018; 34(16):2781–7. Epub 2018/04/05. https:// doi.org/10.1093/bioinformatics/bty185 PMID: 29617937; PubMed Central PMCID: PMC6084588.
- Weissbrod O, Hormozdiari F, Benner C, Cui R, Ulirsch J, Gazal S, et al. Functionally informed fine-mapping and polygenic localization of complex trait heritability. Nat Genet. 2020; 52(12):1355–63. Epub 20201116. https://doi.org/10.1038/s41588-020-00735-5 PMID: <u>33199916</u>; PubMed Central PMCID: PMC7710571.
- Weissbrod O, Kanai M, Shi H, Gazal S, Peyrot WJ, Khera AV, et al. Leveraging fine-mapping and multipopulation training data to improve cross-population polygenic risk scores. Nat Genet. 2022; 54 (4):450–8. Epub 20220407. https://doi.org/10.1038/s41588-022-01036-9 PMID: 35393596; PubMed Central PMCID: PMC9009299.

- Zhang Q, Prive F, Vilhjalmsson B, Speed D. Improved genetic prediction of complex traits from individual-level data or summary statistics. Nat Commun. 2021; 12(1):4192. Epub 2021/07/09. https://doi.org/ 10.1038/s41467-021-24485-y PMID: 34234142; PubMed Central PMCID: PMC8263809.
- Lloyd-Jones LR, Zeng J, Sidorenko J, Yengo L, Moser G, Kemper KE, et al. Improved polygenic prediction by Bayesian multiple regression on summary statistics. Nat Commun. 2019; 10(1):5086. Epub 2019/11/11. https://doi.org/10.1038/s41467-019-12653-0 PMID: <u>31704910</u>; PubMed Central PMCID: PMC6841727.
- Marquez-Luna C, Gazal S, Loh PR, Kim SS, Furlotte N, Auton A, et al. Incorporating functional priors improves polygenic prediction accuracy in UK Biobank and 23andMe data sets. Nat Commun. 2021; 12 (1):6052. Epub 2021/10/20. https://doi.org/10.1038/s41467-021-25171-9 PMID: 34663819; PubMed Central PMCID: PMC8523709.
- 25. Hu Y, Lu Q, Powles R, Yao X, Yang C, Fang F, et al. Leveraging functional annotations in genetic risk prediction for human complex diseases. PLoS Comput Biol. 2017; 13(6):e1005589. Epub 2017/06/09. https://doi.org/10.1371/journal.pcbi.1005589 PMID: 28594818; PubMed Central PMCID: PMC5481142.
- Sohail M, Maier RM, Ganna A, Bloemendal A, Martin AR, Turchin MC, et al. Polygenic adaptation on height is overestimated due to uncorrected stratification in genome-wide association studies. Elife. 2019; 8. Epub 2019/03/22. <u>https://doi.org/10.7554/eLife.39702</u> PMID: <u>30895926</u>; PubMed Central PMCID: PMC6428571.
- Berg JJ, Harpak A, Sinnott-Armstrong N, Joergensen AM, Mostafavi H, Field Y, et al. Reduced signal for polygenic adaptation of height in UK Biobank. Elife. 2019; 8. Epub 2019/03/22. <u>https://doi.org/10.7554/eLife.39725</u> PMID: 30895923; PubMed Central PMCID: PMC6428572.
- Kerminen S, Martin AR, Koskela J, Ruotsalainen SE, Havulinna AS, Surakka I, et al. Geographic Variation and Bias in the Polygenic Scores of Complex Diseases and Traits in Finland. Am J Hum Genet. 2019; 104(6):1169–81. Epub 2019/06/04. https://doi.org/10.1016/j.ajhg.2019.05.001 PMID: 31155286; PubMed Central PMCID: PMC6562021.
- Duncan L, Shen H, Gelaye B, Meijsen J, Ressler K, Feldman M, et al. Analysis of polygenic risk score usage and performance in diverse human populations. Nat Commun. 2019; 10(1):3328. Epub 2019/07/ 28. https://doi.org/10.1038/s41467-019-11112-0 PMID: 31346163; PubMed Central PMCID: PMC6658471.
- Dudbridge F, Newcombe PJ. Accuracy of Gene Scores when Pruning Markers by Linkage Disequilibrium. Hum Hered. 2015; 80(4):178–86. Epub 2015/01/01. <u>https://doi.org/10.1159/000446581</u> PMID: 27576758.
- Hou K, Ding Y, Xu Z, Wu Y, Bhattacharya A, Mester R, et al. Causal effects on complex traits are similar for common variants across segments of different continental ancestries within admixed individuals. Nat Genet. 2023; 55(4):549–58. Epub 20230320. https://doi.org/10.1038/s41588-023-01338-6 PMID: 36941441.
- Saitou M, Dahl A, Wang Q, Liu X. Allele frequency differences of causal variants have a major impact on low cross-ancestry portability of PRS. medRxiv. 2022:2022.10.21.22281371. <u>https://doi.org/10. 1101/2022.10.21.22281371</u>
- Wang Y, Guo J, Ni G, Yang J, Visscher PM, Yengo L. Theoretical and empirical quantification of the accuracy of polygenic scores in ancestry divergent populations. Nat Commun. 2020; 11(1):3865. Epub 2020/08/02. https://doi.org/10.1038/s41467-020-17719-y PMID: 32737319; PubMed Central PMCID: PMC7395791.
- Gurdasani D, Barroso I, Zeggini E, Sandhu MS. Genomics of disease risk in globally diverse populations. Nat Rev Genet. 2019; 20(9):520–35. Epub 2019/06/27. https://doi.org/10.1038/s41576-019-0144-0 PMID: 31235872.
- Martin AR, Kanai M, Kamatani Y, Okada Y, Neale BM, Daly MJ. Clinical use of current polygenic risk scores may exacerbate health disparities. Nat Genet. 2019; 51(4):584–91. Epub 2019/03/31. https:// doi.org/10.1038/s41588-019-0379-x PMID: 30926966; PubMed Central PMCID: PMC6563838.
- Shi H, Gazal S, Kanai M, Koch EM, Schoech AP, Siewert KM, et al. Population-specific causal disease effect sizes in functionally important regions impacted by selection. Nat Commun. 2021; 12(1):1098. Epub 2021/02/19. https://doi.org/10.1038/s41467-021-21286-1 PMID: 33597505; PubMed Central PMCID: PMC7889654.
- Wojcik GL, Graff M, Nishimura KK, Tao R, Haessler J, Gignoux CR, et al. Genetic analyses of diverse populations improves discovery for complex traits. Nature. 2019; 570(7762):514–8. Epub 2019/06/21. https://doi.org/10.1038/s41586-019-1310-4 PMID: 31217584; PubMed Central PMCID: PMC6785182.
- Brown BC, Asian Genetic Epidemiology Network Type 2 Diabetes C, Ye CJ, Price AL, Zaitlen N. Transethnic Genetic-Correlation Estimates from Summary Statistics. Am J Hum Genet. 2016; 99(1):76–88.

Epub 2016/06/21. https://doi.org/10.1016/j.ajhg.2016.05.001 PMID: 27321947; PubMed Central PMCID: PMC5005434.

- Galinsky KJ, Reshef YA, Finucane HK, Loh PR, Zaitlen N, Patterson NJ, et al. Estimating cross-population genetic correlations of causal effect sizes. Genet Epidemiol. 2019; 43(2):180–8. Epub 2018/11/27. https://doi.org/10.1002/gepi.22173 PMID: 30474154; PubMed Central PMCID: PMC6375794.
- 40. Hivert V, Sidorenko J, Rohart F, Goddard ME, Yang J, Wray NR, et al. Estimation of non-additive genetic variance in human complex traits from a large sample of unrelated individuals. Am J Hum Genet. 2021; 108(5):786–98. Epub 2021/04/04. https://doi.org/10.1016/j.ajhg.2021.02.014 PMID: 33811805; PubMed Central PMCID: PMC8205999.
- Dahl A, Nguyen K, Cai N, Gandal MJ, Flint J, Zaitlen N. A Robust Method Uncovers Significant Context-Specific Heritability in Diverse Complex Traits. Am J Hum Genet. 2020; 106(1):71–91. Epub 2020/01/ 07. https://doi.org/10.1016/j.ajhg.2019.11.015 PMID: <u>31901249</u>; PubMed Central PMCID: PMC7042488.
- Blanc J, Berg JJ. Testing for differences in polygenic scores in the presence of confounding. bioRxiv. 2023. Epub 20230822. https://doi.org/10.1101/2023.03.12.532301 PMID: 36993707; PubMed Central PMCID: PMC10055004.